

Citation: Zinck CB, Raveendram Thampy P, Uhlemann E-ME, Adam H, Wachter J, Suchan D, et al. (2023) Variation among strains of *Borrelia burgdorferi* in host tissue abundance and lifetime transmission determine the population strain structure in nature. PLoS Pathog 19(8): e1011572. https://doi.org/10.1371/journal.ppat.1011572

Editor: D. Scott Samuels, University of Montana, UNITED STATES

Received: April 7, 2023

Accepted: July 23, 2023

Published: August 22, 2023

Copyright: © 2023 Zinck et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data presented in this manuscript can be accessed at Dryad. https:// doi.org/10.5061/dryad.3r2280gnh.

Funding: This study was supported by grants awarded to MJV: a Discovery Grant from the Natural Sciences and Engineering Research Council of Canada (NSERC; RGPIN-2019-04483; <u>www.nserc-crsng.gc.ca</u>), an Establishment Grant from the Saskatchewan Health Research Foundation (SHRF; 4583; <u>www.shrf.ca</u>), a singleRESEARCH ARTICLE

Variation among strains of *Borrelia burgdorferi* in host tissue abundance and lifetime transmission determine the population strain structure in nature

Christopher B. Zinck¹*, Prasobh Raveendram Thampy¹, Eva-Maria E. Uhlemann¹, Hesham Adam¹, Jenny Wachter², Danae Suchan^{3,4}, Andrew D. S. Cameron^{3,4}, Ryan O. M. Rego^{5,6}, Dustin Brisson⁷, Catherine Bouchard^{8,9}, Nicholas H. Ogden^{8,9,10}, Maarten J. Voordouw¹*

1 Department of Veterinary Microbiology, Western College of Veterinary Medicine, University of Saskatchewan, Saskatoon, Saskatchewan, Canada, 2 Vaccine and Infectious Disease Organization, University of Saskatchewan, Saskatoon, Saskatchewan, Canada, 3 Institute for Microbial Systems and Society, Faculty of Science, University of Regina, Regina, Saskatchewan, Canada, 4 Department of Biology, University of Regina, Regina, Saskatchewan, Canada, 5 Biology Centre, Institute of Parasitology, Czech Academy of Sciences, České Budějovice, Czech Republic, 6 Faculty of Science, University of South Bohemia, České Budějovice, Czech Republic, 7 Department of Biology, University of Pennsylvania, Philadelphia, Pennsylvania, USA, 8 Public Health Risk Sciences, National Microbiology Laboratory, Public Health Agency of Canada, St Hyacinthe, Quebec, Canada, 9 Groupe de recherche en épidémiologie des zoonoses et santé publique (GREZOSP), Faculté de Médecine Vétérinaire, Université de Montréal, Montreal, Canada, 10 Centre de recherche en santé publique (CReSP), Université de Montréal, Montreal, QC, Canada

* chris.zinck@usask.ca (CBZ); maarten.voordouw@usask.ca (MJV)

Abstract

Pathogen life history theory assumes a positive relationship between pathogen load in host tissues and pathogen transmission. Empirical evidence for this relationship is surprisingly rare due to the difficulty of measuring transmission for many pathogens. The comparative method, where a common host is experimentally infected with a set of pathogen strains, is a powerful approach for investigating the relationships between pathogen load and transmission. The validity of such experimental estimates of strain-specific transmission is greatly enhanced if they can predict the pathogen population strain structure in nature. Borrelia burgdorferi is a multi-strain, tick-borne spirochete that causes Lyme disease in North America. This study used 11 field-collected strains of B. burgdorferi, a rodent host (Mus musculus, C3H/HeJ) and its tick vector (Ixodes scapularis) to determine the relationship between pathogen load in host tissues and lifetime host-to-tick transmission (HTT). Mice were experimentally infected via tick bite with 1 of 11 strains. Lifetime HTT was measured by infesting mice with I. scapularis larval ticks on 3 separate occasions. The prevalence and abundance of the strains in the mouse tissues and the ticks were determined by qPCR. We used published databases to obtain estimates of the frequencies of these strains in wild I. scapularis tick populations. Spirochete loads in ticks and lifetime HTT varied significantly among the 11 strains of B. burgdorferi. Strains with higher spirochete loads in the host tissues were more likely to infect feeding larval ticks, which molted into nymphal ticks that had a higher probability of B. burgdorferi infection (i.e., higher HTT). Our laboratory-based estimates of lifetime

source contract from the Public Health Agency of Canada (4500428376; www.canada.ca/en/publichealth.html), a Venture Grant from the Canadian Lyme Disease Foundation (canlyme.com), and a Wildlife Health Research Fund Grant from the Western College of Veterinary Medicine (wcvm. usask.ca). A Dean's Scholarship from the College of Graduate and Postdoctoral Studies at the University of Saskatchewan provided the stipend and covered tuition for CBZ. The NSERC and SHRF grants provided a top-up salary for PRT. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors declare that they have no competing interests.

HTT were predictive of the frequencies of these strains in wild *I. scapularis* populations. For *B. burgdorferi*, the strains that establish high abundance in host tissues and that have high lifetime transmission are the strains that are most common in nature.

Author summary

For many pathogens, the benefit of higher abundance in host tissues is better transmission to uninfected hosts. Evidence for this relationship is rare because measuring pathogen transmission is difficult. Experimental infection with different pathogen strains in the same animal host allows scientists to test the relationship between abundance in host tissues versus transmission. We performed such a study using the tick-borne bacterium *Borrelia burgdorferi*, which causes Lyme disease in humans in North America. Laboratory mice were infected with 1 of 11 different strains of *B. burgdorferi* and ticks were fed on these mice to measure host-to-tick transmission. We determined the presence and abundance of *B. burgdorferi* in the mouse tissues and in the ticks using basic molecular methods. We found that strains with higher abundance in the mouse tissues had higher lifetime transmission to feeding ticks. We used published databases to obtain approximate estimates of the frequencies of our strains in nature. We found that strains that had higher transmission in the laboratory were also more common in nature. Studies that compare strain fitness between the lab versus nature can help scientists understand the evolution of multi-strain pathogens in response to human control measures.

Introduction

Pathogen evolution is determined by the trade-offs between key life history traits including replication, transmission, persistence (*i.e.*, avoidance of immune clearance by the host), and virulence [1-6]. Pathogen strains that have higher within-host replication (*i.e.*, establish higher abundance in host tissues) are expected to have higher transmission success and/or lower clearance by the host, but they may reduce the survival of the host (*i.e.*, have higher virulence), which shortens the duration of the infection. Pathogens are therefore expected to evolve intermediate virulence, which balances the benefits of higher transmission with the costs of a shorter lifespan of infection [6-10]. Life history theory assumes that variation in these key pathogen traits and their trade-offs are determined at least in part by pathogen genetics [1-6]. A powerful approach for demonstrating the genetic basis of trade-offs is the comparative method, which requires measurement of pathogen traits across a set of genetically distinct pathogen clones or strains in a common host and a common environment [4,8,11,12].

Numerous empirical studies have reported the expected positive relationship between pathogen abundance (or pathogen load) in the host tissues and pathogen transmission [7–10,13– 15]. However, upon closer inspection, only one study demonstrates the presence of a genetic correlation between these two pathogen traits by measuring them across a set of parasite clones in experimentally infected hosts [12]. Studies on human pathogens have demonstrated the expected positive relationship but cannot control for host or environmental factors [7,8]. Some studies using arthropod hosts as model systems have measured genetic variation among pathogen strains in load or virulence but not in transmission [9,10]. Thus, there are surprisingly few examples of a genetic correlation between pathogen load in host tissues and pathogen transmission and recent reviews have highlighted the need for more such studies [4,6,15]. Studies that measure pathogen life history traits are typically conducted in the lab under carefully controlled conditions. An open question is whether these lab-based estimates of strain fitness are relevant to the real world. The revolution in next generation sequencing methods has allowed scientists to document pathogen strain diversity in host populations with great accuracy and precision [16–20]. All else being equal, the frequency of a strain in the pathogen population is a measure of its fitness at a point in time; common strains are expected to have higher fitness compared to rare strains. Thus, one way to validate laboratory-based estimates of pathogen strain fitness would be to test whether they can explain variation in frequencies among strains in the real world [18,20]. Such validation would increase our confidence that lab-based estimates of the genetic variation underlying pathogen life history traits are relevant to their evolution in nature.

Lyme borreliosis is the most common vector-borne disease in Europe and North America [21] and is caused by tick-borne spirochetes belonging to the *Borrelia burgdorferi* sensu lato (sl) genospecies complex. The three most important *Borrelia* genospecies are *B. afzelii* and *B. garinii* in Europe, and *B. burgdorferi* sensu stricto (hereafter *B. burgdorferi*) in North America [22]. These genospecies are transmitted among vertebrate hosts through the bite of *Ixodes* ticks, which include *I. scapularis* in North America and *I. ricinus* in Europe. *Ixodes* ticks consist of three feeding developmental stages: larva, nymph, and adult. Larval ticks acquire spirochetes after feeding on infected hosts and moult into infected nymphs, which infect the next generation of vertebrate hosts. Adult female ticks feed on deer that are incompetent for *B. burgdorferi* sl pathogens [23,24]. In summary, the two transmission steps that maintain *B. burgdorferi* sl in nature are from infected vertebrate hosts to naïve larvae and from infected nymphs to naïve vertebrate hosts.

Each *B. burgdorferi* sl genospecies comprises multiple strains that coexist in natural populations. These strains can vary in their life history traits, such as replication or abundance in host tissues [25-29] or host-to-tick transmission (HTT) [30-37]. However, it is generally thought that *B. burgdorferi* sl does not elicit pathology in important rodent reservoir hosts like the white-footed mouse (*Peromyscus leucopus*), which in most studies (but not all), appears to be tolerant of infection [38-41]. Few studies have investigated covariation between life history traits for any *B. burgdorferi* sl pathogen. A study of two strains of *B. afzelii* found that the strain abundance in host tissues was correlated with HTT [26]; but the sample size (n = 2 strains) was not sufficient to test for genetic correlations between pathogen life history traits. Other studies have found that lab-based estimates of spirochete load in host tissues or lab-based estimates of transmission are associated with strain frequencies in nature [18,25,42]. However, no study has investigated the relationships between host tissue spirochete load, lifetime HTT, and frequency in nature across a set of *B. burgdorferi* sl strains. Such studies enhance our understanding of the ecological and evolutionary forces that shape the current strain structure, and which strains are most likely to be of public health significance.

In this study, we investigated the variation in host-to-tick transmission (HTT) among 11 different strains of *B. burgdorferi*. We have previously shown that there is significant variation in host tissue abundance among these strains [42]. The present study had three objectives. First, characterize the variation in lifetime HTT among 11 strains of *B. burgdorferi*. Second, quantify the relationship between host tissue spirochete load (replication) and lifetime HTT for 11 strains of *B. burgdorferi*. Third, determine whether our laboratory estimates of strainspecific lifetime HTT are correlated with strain frequencies in nature. We hypothesize that strains with higher tissue spirochete load will have a greater lifetime HTT. Finally, we predict that the strains with a greater lifetime HTT will be more common in nature.

Materials and methods

Ethics statement

The use of mice and all procedures in this study were reviewed and approved by the Animal Research Ethics Board (AREB), which is part of the University Animal Care Committee (UACC) at the University of Saskatchewan. The AREB-assigned animal use protocol number for our study was 20190012. The UACC follows the animal experimentation guidelines of the Canadian Council on Animal Care (CCAC).

Strains, ticks, and mice

For this study, 11 low passage strains of *B. burgdorferi* were obtained from the isolate collection of the Public Health Agency of Canada (PHAC; Table 1). To minimize plasmid loss during culture, the *B. burgdorferi* strains had undergone only four passages from the time of isolation from field-collected *I. scapularis* ticks to needle inoculation into our mice. Initially, 12 strains were used, however, one failed to infect mice and is not included here [42]. These strains were originally isolated from wild *I. scapularis* ticks collected from field sites in Canada [43]. Strains were whole genome sequenced and their multi-locus sequence types (MLSTs) and *ospC* types were determined by PHAC [43]. The strains were selected by maximizing MLST diversity while also considering the *ospC* type and geographic region of origin. We selected 5 strains from Eastern Canada (Nova Scotia), and 6 from Midwestern Canada (Manitoba and western Ontario) because previous work found substantial differences in strain diversity between these two regions [44–46]. All strains were validated for the presence of plasmids that are necessary for the natural transmission cycle of *B. burgdorferi* [42].

Laboratory-reared, specific-pathogen-free *I. scapularis* larvae were obtained from the National Tick Research and Education Resource at Oklahoma State University. This colony is

i								
PubMLST Strain ID	NML Strain ID	MLST	ospC type	Region	Town (Province)	Infected mice	Infected larvae	Infected nymphs
2388	10-2	32	C3	Midwest	Roseau River (MB)	8/8	62/87	202/234
2392	22-2	55	А	Midwest	Buffalo Point (MB)	8/8	53/83	176/226
2038	54	741	I	Midwest	Buffalo Point (MB)	8/8	33/93	155/230
2439	57-1	29	A3	Midwest	Buffalo Point (MB)	8/8	70/90	212/230
2443	66	237	J	Midwest	Buffalo Point (MB)	7/8	27/77	114/210
2399	126	43	N	Midwest	Big Grassy (ON)	8/8	58/90	210/236
2413	150	19	Е	East	Lunenberg (NS)	8/8	61/94	205/239
2416	167	12	М	East	Shelburne (NS)	8/8	66/93	200/240
2418	174	37	Т	East	Lunenberg (NS)	5/8	47/58	148/150
2420	178-1 ^a	3	К	East	Lunenberg (NS)	8/8	85/89	233/236
2429	198	3	К	East	Bedford (NS)	8/8	83/96	229/240
	Total					84/88	645/950 ^c	2,084/2,471 ^c
	Uninfected ^b	NA	NA	NA		0/20	0/200	0/558

Table 1. The genetic identity and location of origin are shown for the 11 strains of *B. burgdorferi* used in the study. Strain identity numbers (Bb16-) are from the reference publication [43]. Counts of larvae and nymphs are only for the subset of infected mice as mice that failed to become infected were excluded from analysis.

^a This strain was originally intended to be 178–2 but turned out to be 178–1 (Tyler et al., 2018)[43], thus we have two isolates of the same strain in this study (MLST 3, *ospC* K).

^b Uninfected control mice were fed upon by uninfected *I. scapularis* nymphs. As expected, all the ticks that fed on the uninfected control mice tested negative for *B. burgdorferi*.

^c The larvae and nymphs from the 4 mice that failed to become infected following the nymphal challenge are not included in these numbers

https://doi.org/10.1371/journal.ppat.1011572.t001

regularly outbred with wild *I. scapularis* ticks to prevent inbreeding and is screened for multiple tick-borne pathogens including *B. burgdorferi*. Six batches of larvae were purchased over the course of the study (2 temporal blocks x 3 larval infestations per block). The larvae for each infestation came from egg clutches of different adult female ticks.

Specific-pathogen-free, 6 to 8-week-old, female and male *Mus musculus* C3H/HeJ mice were purchased from The Jackson Laboratory and used as the reservoir host in this study. This strain of mouse has been used in numerous experimental infection studies with *B. burgdorferi* [28,29,47]. *Mus musculus* mice are not natural reservoir hosts for *B. burgdorferi*. The main reasons for working with this non-natural reservoir host were availability, convenience, cost, and genetic similarity of individual mice. While strain C3H/HeJ possesses a mutation in Toll-like receptor 4, which makes it highly susceptible to Gram-negative bacteria compared to the wild-type C3H/HeN, previous studies have found that C3H/HeJ mice are no different than wild type in their response to infection with *B. burgdorferi* [48,49].

Creation of *I. scapularis* nymphs infected with one of the 11 strains of *B. burgdorferi*

The *I. scapularis* nymphs were infected with one of the 11 strains as described previously [42]. Briefly, each strain was cultured from frozen stock in Barbour-Stonner-Kelly-H (BSK-H) medium (Millipore Sigma, Oakville, Canada) for 9–14 days. C3H/HeJ mice were then needle-inoculated with single strains (n = 2–4 mice per strain). At 2–4 weeks post-infection (PI), mice were infested with naïve *I. scapularis* larvae that were left to feed to repletion following our infestation procedure (see below). These larvae were maintained at room temperature in humidity chambers to moult into nymphs.

Infecting mice via nymphal tick bite with one of the 11 strains of *B*. *burgdorferi*

A total of 108 mice (54 female, 54 male) were used in this study. For each of the 11 strains, 8 mice (4 female, 4 male) were infected, with an additional 20 mice (10 female, 10 male) being uninfected controls. To mimic a natural infection cycle, experimental mice were infected by allowing three putatively infected *I. scapularis* nymphs to feed to repletion as previously described [42]. Control mice were infested with 3 uninfected *I. scapularis* nymphs. To manage the workload, the study was run in two orthogonal temporal blocks (A, B) processed ~6 months apart; each block contained half the experimental mice and half the control mice. To confirm the infectious challenge, replete nymphs were recovered and tested for the presence of *B. burgdorferi* by qPCR targeting the 23S rRNA gene of *B. burgdorferi* (see below).

Infestations of mice with I. scapularis larvae

At days 30, 60, and 90 PI, mice were infested with 50–100 naïve *I. scapularis* larval ticks that were allowed to feed to repletion. All mice were anaesthetized for ~20 min with xylazine and ketamine for each infestation, and larvae were brushed onto their fur. Mice were housed in a cage setup that facilitated the collection of engorged larvae as described previously [27]. For each mouse and infestation, 5 engorged larvae were frozen immediately at -80°C following recovery (no engorged larvae were collected for the first infestation of the mice in block A). The remaining engorged larvae were kept at high humidity (~99%) and room temperature and were allowed to moult into nymphs. Four weeks after the engorged larvae moulted into nymphs, a sample of 10 nymphs per mouse per infestation were frozen at -80°C for DNA

extraction. These nymphs will hereafter be referred to as unfed nymphs or flat nymphs to indicate that they have not taken their nymphal blood meal.

Collection of ear biopsies and tissue samples from mice at necropsy

The sampling of the ear biopsies and necropsy tissue samples from the mice in this study was previously described in [42]. Ear biopsies (2 mm in diameter) were collected from the right ear on days 29, 59, and 89 PI, which were the days before the mice were infested with *I. scapularis* larvae. The mice were euthanized on day 97 PI, and we dissected and collected 7 necropsy tissues: (1) ventral skin, (2) left ear, (3) right ear, (4) heart, (5) bladder, (6) right rear tibiotarsal joint, and (7) kidney. Mouse necropsy tissue samples were frozen at -80°C prior to DNA extraction.

DNA extraction of whole I. scapularis ticks

For each mouse, there were 45 ticks (3 infestations * (5 engorged larvae + 10 flat nymphs moulted from larvae fed on the mouse) per infestation) for a total of 4,860 ticks (108 mice x 45 ticks). Tick extractions were done as previously described [27]. Ticks were homogenized using bead beating with the Qiagen TissueLyser II. Homogenized ticks were digested with proteinase K overnight at 56°C in a shaking incubator at 400 rpm. DNA was extracted from the homogenized ticks using the Qiagen DNEasy 96-well plate extraction kit following the manufacturer's instructions. The DNA extractions of the ticks were eluted with 65 μ L of elution buffer and stored at -80°C prior to use in qPCR.

DNA extraction of mouse ear biopsies and necropsy tissues

The homogenization and DNA extraction of the mouse ear biopsies and necropsy tissue samples in this study were previously described in [42]. Briefly, a total of 324 ear biopsies (108 mice x 3 ear biopsies per mouse) and 756 mouse tissue samples from necropsies were processed (108 mice x 7 tissue samples per mouse); the 7 necropsy samples included (1) ventral skin, (2) left ear, (3) right ear, (4) heart, (5) bladder, (6) right rear tibiotarsal joint, and (7) kidney. Whole ear biopsies and the necropsy samples (~20 mg of tissue) were homogenized using 3.6-mm stainless steel beads with the Qiagen TissueLyser II or a micro pestle. Homogenized tissue samples were digested with proteinase K overnight at 56°C in a shaking incubator at 400 rpm. DNA was extracted from homogenized tissue samples using the Qiagen DNEasy Blood and Tissue kit individual spin columns following the manufacturer's instructions. The DNA extractions of the ear biopsies and necropsy tissues were eluted with 55 μ L and 100 μ L of elution buffer, respectively, and stored at -80°C prior to use in qPCR.

Detection of B. burgdorferi in I. scapularis ticks by qPCR

To test for the presence of *B. burgdorferi* in experimental samples, a qPCR targeting the 23S *rRNA* intergenic spacer of *B. burgdorferi* was used [50]. Every plate was run with a synthetic gene fragment to use as a standard [42]. A strict lower detection limit of 1 gene copy/ μ L was used as a cut-off based on the lower detection limit of the standards. To generate copy numbers, a single regression was done on multiple runs of standards (range: 1–10⁵ copies/ μ L, n = 81) so that the same equation was used for every transformation [42]. As whole ticks were used in every DNA extraction, values are reported as the log₁₀ 23S *rRNA* copies per tick.

In comparison to nymphs, replete larvae have a lower abundance of *B. burgdorferi* and the presence of mammalian blood can inhibit amplification of DNA [51,52]. For this reason, a qPCR targeting the *I. scapularis* calreticulin gene was developed (**see ESM Section 1**) to

determine DNA extraction success by using a tick housekeeping gene present in all samples. As with the *23S rRNA* qPCR, a synthetic gene fragment was designed to use as a standard (**see ESM Section 2**).

Detection of B. burgdorferi in mouse tissues by qPCR

The abundance of *B. burgdorferi* in the mouse ear biopsies and necropsy tissue samples was also estimated using 23S rRNA qPCR and was standardized relative to the number of mouse *Beta-actin* gene copies, as previously described [42].

Statistical methods

Infection prevalence and spirochete load of mouse tissues. The infection prevalence and the spirochete load of the mouse ear biopsies and the 7 mouse tissue samples were analyzed as described in ESM Section 3 (see also [42]). The infection prevalence refers to whether the mouse tissue is uninfected (0) or infected (1). In the present study, the spirochete load in the mouse tissues was calculated as log10[(23S rRNA gene copies/10^6 Beta-actin copies) + 1]. The advantage of this log[X + 1] transformation is that uninfected tissues are defined and have a value of 0. The mouse tissue spirochete loads were scaled to z-scores with mean of 0 and a variance of 1 (hereafter referred to as the z-score spirochete loads). The advantage of this scaling is that each tissue has the same weight in estimates that are averaged across tissues.

Overview of transmission measures: LIP, LSL, NIP, and NSL. We analyzed 4 measures of transmission of *B. burgdorferi*: (1) proportion of engorged larvae infected with *B. burgdorferi*, hereafter larval infection prevalence (LIP), (2) abundance of *B. burgdorferi* in the subset of infected larvae, hereafter larval spirochete load (LSL) (3) proportion of unfed nymphs infected with *B. burgdorferi*, hereafter nymphal infection prevalence (NIP), and (4) abundance of *B. burgdorferi* in the subset of infected unfed nymphs, hereafter nymphal spirochete load (NSL). The details of these analyses are in **ESM Sections 4, 5, 6, and 7**.

Analysis of *B. burgdorferi* infection prevalence in larvae and nymphs. For larvae and nymphs, the presence of *B. burgdorferi* infection was a binomial variable (0 = uninfected, 1 = infected). The LIP and the NIP were analyzed by generalized linear mixed-effects models (GLMMs) with binomial errors. For both analyses, the fixed factors included strain (11 levels; see <u>Table 1</u>), mouse sex (2 levels: female, male), infestation (3 levels: days 30, 60, 90 PI), and their interactions. Temporal block (2 levels: A, B) was included as a main effect. Mouse identity was included as a random factor. Analyses were restricted to ticks from the subset of 84 mice that became infected following nymphal challenge (see below). To generate the final models, interactions were sequentially removed if they were not significant and/or if their inclusion prevented the GLMM from converging on precise parameter estimates. Factor significance was estimated using type II Wald tests, post-hoc analyses were performed and estimated marginal means (EMMs) were calculated.

Analysis of *B. burgdorferi* **abundance in infected larvae and infected nymphs.** For the subset of infected larvae and nymphs, analyses were run using linear mixed-effects models (LMMs). For both analyses, the fixed factors included strain, mouse sex, infestation, and their interactions. Temporal block was included as a main effect. Mouse identity was included as a random factor. For the larvae only, calreticulin copies per tick were also included as a covariate to control for random variation in the efficiency of DNA extraction. Non-significant interactions were sequentially removed to generate the final models. Post-hoc analyses and EMMs were performed as mentioned above.

Relationship between the mouse tissue spirochete loads and NIP. We wanted to determine which mouse tissues were most strongly associated with variation in the NIP (see ESM **Section 8 for details**). There were 8 mouse tissues: ear biopsies that were sampled the day before the larval infestation (i.e., days 29, 59, and 89 PI) and 7 mouse tissues that were sampled at necropsy (day 97 PI). Each mouse tissue had an estimate of spirochete load, which was calculated as log10[(*23S rRNA* gene copies/10^6 *Beta-actin* copies) + 1]. The mouse tissue spirochete loads were scaled to z-scores with a mean of 0 and a variance of 1 (hereafter referred to as the z-score spirochete loads).

We used AIC-based model selection to compare 20 GLMMs with binomial errors; each model analyzed the NIP as a function of a different mouse tissue variable (**see ESM Section 8 for details**), as well as the fixed factors of mouse sex, infestation, and relevant interactions. Mouse identity was included as a random factor. Models were ranked according to their corrected Akaike Information Criterion (AICc) where the best models have the lowest AICc scores. The parameter estimates of the best models were analyzed in more detail.

Correlation matrix of mouse tissue spirochete loads and *B. burgdorferi* transmission **variables.** We wanted to visualize the pairwise correlation matrix for our eight estimates of tissue spirochete load: (1) ear biopsy, (2) ventral skin, (3) left ear, (4) right ear, (5) heart, (6) bladder, (7) kidney, (8) and right rear tibiotarsal joint, and our four estimates of host-to-tick transmission: (9) larval infection prevalence (LIP), (10) larval spirochete load (LSL), (11) nymphal infection prevalence (NIP), and (12) nymphal spirochete load (NSL). The genetic correlation matrix was calculated across the 11 *B. burgdorferi* strains, whereas the phenotypic correlation matrix was calculated across the 84 infected mice (**see ESM Section 9**).

Frequencies of *B. burgdorferi* strains in *I. scapularis* tick populations in North America. We wanted to use the frequencies of our *B. burgdorferi* strains in *I. scapularis* tick populations in North America as a measure of strain fitness. Two studies by Travinsky et al [53] and Ogden et al [46] determined the frequencies of *B. burgdorferi* MLSTs in *I. scapularis* ticks collected in the USA and Canada. Travinsky et al [53] classified MLSTs as belonging to the northeastern or north central (hereafter midwestern) USA. Although Ogden et al [46] sampled *I. scapularis* ticks in Canada, they similarly classified the MLSTs as being the same as, or clustering with, MLSTs that had been previously detected in the northeastern or midwestern USA. Our study contained 11 *B. burgdorferi* strains with 10 unique MLSTs (strain 178–1 and strain 198 both have MLST 3). We obtained frequency estimates for our strains by matching their MLSTs to the MLSTs (without regard to *ospC* type) in these published data sets (**see ESM Section 10 for details**).

MLSTs from regions not relevant to our study (e.g., northern California and Europe) were excluded. After this step, Travinsky et al [53] identified 35 different MLSTs in 654 *I. scapularis* ticks and Ogden et al [46] identified 40 different MLSTs in 166 *I. scapularis* ticks. Together, these two studies contained 8 of the 10 unique MLSTs and 9 of the 11 strains in the present study. The two strains that were not detected in either study, strains Bb16-66 and Bb16-54 with MLSTs 237 and 741, were assigned counts of zero. Strains Bb16-10-2 and Bb16-57-1 had matching MLSTs in the study by Travinsky et al [53] but different *ospC* types. For each study, we counted the number of *I. scapularis* ticks infected with each of the 10 unique MLSTs of interest. To obtain the best estimate of the MLST-specific frequencies in *I. scapularis* populations, the counts were combined for the two studies.

For the two studies combined, the 10 unique MLSTs in our study were detected in 144 and 109 ticks in northeastern and midwestern North America, respectively (total n = 253 ticks). For the two studies combined, 40 MLSTs were identified in 482 and 338 ticks in northeastern and midwestern North America, respectively (total n = 820 ticks). These counts were converted to four different strain-specific frequencies of which two are presented here (**see ESM Sections 10 and 11**). For frequency 2 (%), the counts for the 10 MLSTs were divided by the number of ticks when the two geographic regions were combined (n = 820). For frequency 4

(%), the counts for the 4 MLSTs in the northeast and the 6 MLSTs in the Midwest were divided by the number of ticks in each region (n = 482 and n = 338). Thus, frequency 4 accounts for potential differences in sampling effort between geographic regions, whereas frequency 2 does not.

Statistical software

The data analyzed for this manuscript can be accessed at Dryad [54]. We used R version 4.0.4 for all statistical analyses [55]. The list of R packages and R functions we used are given in **ESM** Section 1.

Results

Infestations of mice and recovery and testing of immature *I. scapularis* ticks

For each strain, 5–8 infected mice survived to the end of the study with a total of 84 infected mice surviving (Table 1). Of the 4 mice that did not become infected following nymphal challenge, 3 had been infected with strain Bb16-174 and 1 with Bb16-66 (Table 1). In addition, 20 uninfected control mice survived to the end of the study. For each mouse and each infestation, we froze a maximum of 5 engorged larvae and 10 unfed nymphs. The infection status of these mice was previously determined by performing qPCR on 3 post-infection ear biopsies and 7 tissues collected at necropsy and by performing ELISA on 2 post-infection blood samples per mouse [42]. In the present study, a maximum of 45 immature ticks (15 larvae, 30 nymphs) were tested for their infection status for each mouse. Every infected mouse was found positive for *B. burgdorferi* in mouse tissues, xenodiagnostic ticks, and with respect to the IgG antibody response on a commercial ELISA. No evidence of *B. burgdorferi* infection was found in the 20 control mice, or in the 4 mice that failed to demonstrate infection status of the 84 infected mice, the 4 experimental mice that failed to become infected following nymphal challenge, and the 20 uninfected control mice.

Infection prevalence and spirochete load of mouse tissues

The infection prevalence and the spirochete loads of the 3 ear biopsies and the 7 necropsy tissues are presented in detail in **ESM Section 3** (see also [42]). The infection prevalence in the biopsies from the right ear on days 29, 59, and 89 PI were 98.8%, 100.0%, and 91.7%, respectively. For the necropsy tissues on day 97 PI, the infection prevalence was low in skin-related tissues (52.4% in ventral skin, 38.1% in left ear, and 14.3% in right ear) and was high in internal tissues (92.9% in heart, 97.6% in bladder, and 71.4% in right rear tibiotarsal joint).

The spirochete loads are measured in units of 23S rRNA copies per 10^6 Beta-actin copies. For the biopsies of the right ear, the mean spirochete load decreased by 71.8% from the first to the third biopsy (2878.5 versus 812 units). For the necropsy tissues on day 97 PI, if the right ear is taken as the reference, the mean spirochete load in the kidney, left ear, ventral skin, joint, bladder, and heart is 3.2x, 8.0x, 14.3x, 32.6x, 76.9x, 824.2x, and 1182.9x higher, respectively. Thus, the mean spirochete load in the skin-related organs (ventral skin, ears) was much lower compared to the internal organs (bladder, and heart).

An unexpected result is that the infection prevalence in the right ear decreased from 91.7% in the biopsy on day 89 PI to 14.3% in the necropsy sample on day 97 PI (a 6.4-fold decrease). Similarly, the spirochete load in the right ear decreased from 812 units in the biopsy on day 89 PI to 1.3 units in the necropsy sample on day 97 PI (a 624.6-fold decrease). Of note, all ear

biopsies were collected 1 day before infestation with larvae. However, ear tissue samples collected at necropsy on day 97 PI were collected one week after larvae fed on the mice.

Larval infection prevalence

Each mouse had up to 5 engorged larvae frozen from three separate infestations at days 30, 60, and 90 PI. In total, 950 and 200 engorged larvae were recovered and tested from the 84 infected mice and 20 uninfected control mice, respectively. The infection prevalence in the engorged larvae that fed on the 84 infected mice was 67.9% (645/950) with none of the 200 larvae from the 20 uninfected control mice testing positive (Table 1).

The analysis of the LIP found that strain ($p = 1.346*10^{-15}$) and infestation ($p = 3.359*10^{-5-}$) were significant (see ESM Section 4). The overall LIP varied 2.7-fold among strains (strain Bb16-54: 35.5%; strain Bb16-178-1: 95.5%). The LIP decreased from the first infestation at day 30 PI to the subsequent infestations (30 days PI: 89.7%; 60 days PI: 60.5%; 90 days PI: 64.6%).

Larval spirochete load

The analysis of the LSL for the subset of infected engorged larvae (n = 645) found that strain (p = $5.968*10^{-12}$) and infestation (p = $4.773*10^{-11}$) were both significant (see ESM Section 5). The LSL has units of *23S rRNA* copies/larva and varied 11.9-fold between the lowest and highest abundance strains (strain Bb16-54: 1,317.4 units; strain Bb16-178-1: 15,635.5 units). As with the LIP, the LSL was highest in the first infestation (30 days PI: 10,501.0 units; 60 days PI: 2,115.3 units; 90 days PI: 5,538.3 units).

Nymphal infection prevalence (host-to-tick transmission)

Each mouse had up to 10 unfed, 4-week-old nymphs frozen from the three separate infestations at days 30, 60, and 90 PI. In total, 2,471 and 558 unfed nymphs were tested from the 84 infected mice and 20 uninfected control mice, respectively. The infection prevalence in the unfed nymphs that had fed as larvae on the 84 infected mice was 84.3% (2,084/2,471) with none of the 558 nymphs that had fed as larvae on the 20 uninfected control mice testing positive (Table 1).

The analysis of the NIP found that strain ($p = 1.758*10^{-14}$), mouse sex (p = 0.001), infestation ($p = 2.2*10^{-16}$), and the 2-way interaction between sex and infestation ($p = 1.625*10^{-4}$) were significant (**see ESM Section 6**). For 8 of the 11 strains, the NIP decreased with successive larval infestations, whereas it remained consistently high (above 90%) for 3 strains (Bb16-174, Bb16-178-1, Bb16-198; **Fig 1**). At day 30 PI, the NIP for female and male mice was similar (females: 97.8%; males: 94.8%) but declined more for females by day 90 PI (females: 67.1%; males: 82.6%; **Fig 2**). Thus, the NIP decreased over time in both sexes, but the decrease in NIP was significantly greater for female mice (n = 43), which was higher compared to 80.7% (971/1,203) in female mice (n = 41). The lifetime NIP was higher for male mice compared to female mice for 9 of 11 *B. burgdorferi* strains.

Nymphal spirochete load

The analysis of the NSL in the subset of infected unfed nymphs (n = 2,084) found that strain (p = $2.2*10^{-16}$), infestation (p = $8.914*10^{-12}$), and the 2-way interactions between strain and infestation (p = $7.561*10^{-5}$), and between mouse sex and infestation (p = 0.026) were all significant (see ESM Section 7). The NSL has units of *23S rRNA* copies/nymph and varied 8.6-fold between the lowest and highest abundance strains (strain Bb16-66: 7,707.3 units; strain Bb16-





174: 66,069.3 units). Unlike the LSL, the NSL in the unfed nymphs increased for some strains with successive infestations, whereas it decreased for others. Like the NIP, the NSL in ticks fed on female mice decreased more with successive infestations than the NSL in ticks fed on male mice (Males: 30 days PI: 26,525.2 units; 90 days PI: 19,075.1 units; Females: 30 days PI: 32,587.7 units; 90 days PI: 17,937.3 units).

Relationship between the mouse tissue spirochete loads and NIP

The AIC-based model selection of the 20 GLMMs to determine which mouse tissues influenced the NIP is presented in **ESM Section 8**. The top 5 GLMMs of the NIP had 99.2% of the support. These 5 models included spirochete loads for the (1) heart, (2) bladder, (3) mean of heart and bladder, (4) ear biopsy, and (5) infection prevalence for the heart with 30.7%, 29.7%, 26.7%, 7.5% and 4.7% of the support, respectively. The GLMMs that modelled the NIP as a function of heart spirochete load or ear biopsy spirochete load are discussed below.



Fig 2. Host-to-tick transmission of *B. burgdorferi* from infected male and female C3H/HeJ mice to immature *I. scapularis* ticks decreased over the course of the infection. Male mice had higher lifetime HTT compared to female mice for all 11 strains of *B. burgdorferi*. Lifetime HTT was measured by infesting mice with *I. scapularis* larvae on 3 occasions (days 30, 60, and 90 post-infection), allowing the engorged larvae to moult into nymphs, and testing the nymphs for infection with *B. burgdorferi*. The highly significant 2-way interaction between mouse sex and infestation (p = 0.0002) indicates that the decrease in the HTT over time was larger in the female mice compared to the male mice.

The GLMM that modelled the NIP as a function of heart spirochete load, mouse sex, and infestation is described in **ESM Section 8**. The heart spirochete load had a significant and positive effect on the NIP (slope = 0.751, SE = 0.164, p < 0.001). This result indicates that *B. burg-dorferi* strains that establish higher spirochete loads in the heart tissues have higher host-to-tick transmission to feeding ticks. To visualize the relationship, we graphed the mean lifetime NIP for each strain versus its mean heart spirochete load (**Fig 3**). The lifetime NIP was calculated by summing the infected and uninfected nymphs over all infestations and mice for each of the 11 *B. burgdorferi* strains. There was a strong positive relationship between heart spirochete load and lifetime NIP across the 11 strains of *B. burgdorferi* (**Fig 3**; slope = 0.890, SE = 0.214, p = 0.002).

The GLMM that modelled the NIP as a function of ear biopsy spirochete load, mouse sex, and infestation is described in **ESM Section 8**. A significant 2-way interaction (p = 0.002) indicates that the slope relating NIP to ear biopsy spirochete load differs between infestations. The slope of the relationship between the NIP and the ear biopsy spirochete load was negative for



Fig 3. The mean spirochete load in the mouse heart determines the lifetime host-to-tick transmission (HTT) across the 11 strains of *B. burgdorferi*. The lifetime HTT (also called the lifetime NIP) is the percentage of nymphs infected with *B. burgdorferi*; these nymphs had taken their larval blood meal on the mice at days 30, 60, or 90 PI. The lifetime HTT for each strain was calculated by summing the infected nymphs and uninfected nymphs over all infestations and mice. The hearts were dissected from the euthanized mice at day 97 PI. The spirochete load in the mouse heart was calculated as log10[(*23S rRNA*/ 10⁶ *Beta-actin*) + 1] and was converted to a z-score. A GLM with quasibinomial errors found a strong positive relationship between heart spirochete load and lifetime HTT across the 11 *B. burgdorferi* strains (slope = 0.890, SE = 0.214, p = 0.002). Strains from eastern and midwestern Canada are shown in blue and red, respectively. Labels on datapoints are strain ID numbers.

infestation 1 (slope = -0.790, SE = 0.278, p = 0.004), whereas it was positive for infestation 2 (contrast of slopes = 1.261, SE = 0.407, p = 0.002) and for infestation 3 (contrast of slopes = 1.011, SE = 0.300, p = 0.001). This result indicates that *B. burgdorferi* strains that establish higher spirochete loads in the ear tissues have higher host-to-tick transmission to feeding ticks for infestation 2 at day 60 PI and for infestation 3 at day 90 PI. To visualize the relationship, we graphed the mean lifetime NIP for each strain versus its mean ear biopsy spirochete load (**Fig 4**). There was a strong positive relationship between ear biopsy spirochete load and lifetime NIP across the 11 strains of *B. burgdorferi* (**Fig 4**; slope = 2.357, SE = 0.467, p < 0.001).

Correlation matrix of mouse tissue spirochete loads and *B. burgdorferi* transmission variables

To visualize the pairwise correlations between our 8 mouse tissue spirochete load variables and our 4 *B. burgdorferi* transmission variables, we calculated the genetic and phenotypic correlation matrices. For the genetic correlation matrix, the four estimates of host-to-tick transmission, LIP, LSL, NIP, and NSL, were all positively correlated with each other (Fig 5). Spirochete loads of the ear biopsies, heart, bladder, and kidney were positively correlated with the NIP and the NSL. Spirochete loads of the ear biopsies, heart, and bladder were positively correlated with the LIP and the LSL. The results for the phenotypic correlation matrix were very similar (see ESM Section 9).



Fig 4. The mean spirochete load in the mouse ear biopsy determines the lifetime host-to-tick transmission (HTT) across the 11 strains of *B. burgdorferi*. The lifetime HTT (also called the lifetime NIP) is the percentage of nymphs infected with *B. burgdorferi*; these nymphs had taken their larval blood meal on the mice at days 30, 60, or 90 PI. The lifetime HTT for each strain was calculated by summing the infected nymphs and uninfected nymphs over all infestations and mice. The ear biopsies were sampled on days 30, 60, and 90 PI and were averaged for each mouse and strain. The spirochete load in the mouse ear biopsy was calculated as $log10[(23S rRNA/10^6 Beta-actin) + 1]$ and was converted to a z-score. A GLM with quasibinomial errors found a strong positive relationship between ear biopsy spirochete load and lifetime HTT across the 11 *B. burgdorferi* strains (slope = 2.357, SE = 0.467, p < 0.001). Strains from eastern and midwestern Canada are shown in blue and red, respectively. Labels on datapoints are strain ID numbers.

Relationship between lifetime NIP and frequencies observed in *I. scapularis* ticks in North America across 11 *B. burgdorferi* strains

We tested whether there was a relationship between the lifetime NIP and the MLST frequencies observed in *I. scapularis* ticks in North America across the 11 *B. burgdorferi* strains (**see ESM Sections 10 and 11 for details**). The lifetime NIP was calculated by summing the infected and uninfected nymphs over all infestations and mice for each of the 11 *B. burgdorferi* strains. We analyzed the frequencies that included the other MLSTs when the two regions were combined (i.e., MLST frequency 2), and when the two regions were adjusted for sampling effort (i.e., MLST frequency 4). MLST frequency was analyzed using a GLM with quasibinomial errors (to correct for overdispersion) as a function of geographic region (2 levels: northeastern North America and midwestern North America), strain-specific lifetime NIP, and their interaction.

For MLST frequency 2, geographic region (p = 0.970) and its interaction with lifetime NIP (p = 0.962) were not significant and were therefore removed from the model. Lifetime NIP had a significant effect on MLST frequency 2 (p = 0.001). The slope of the relationship between lifetime NIP and MLST frequency 2 was positive (**Fig 6**; slope = 9.685, SE = 3.724, p = 0.029). Strains with higher lifetime NIP had a higher frequency in *I. scapularis* populations in nature. The results were similar for MLST frequency 4 (**see ESM Section 11 for details**). The slope of the relationship between lifetime NIP and MLST frequency 4 was positive (slope = 8.708, SE = 3.942, p = 0.055). After adjusting for differential sampling effort between the two geographic regions, strains with higher lifetime NIP had a higher frequency in *I. scapularis* populations in nature.



Fig 5. Genetic correlation matrix for 12 variables across the 11 strains of *B. burgdorferi.* The 12 variables include spirochete loads in eight mouse tissues: (1) ear biopsy, (2) ventral skin, (3) left ear, (4) right ear, (5) heart, (6) bladder, (7) kidney, (8) right rear tibiotarsal joint, and four estimates of host-to-tick transmission of *B. burgdorferi*: (9) larval infection prevalence (LIP), (10) larval spirochete load (LSL), (11) nymphal infection prevalence (NIP), and (12) nymphal spirochete load (NSL). The NIP is equivalent to host-to-tick transmission (HTT). The size and color of the circle indicate the absolute value and direction of the Pearson correlation coefficient, respectively. Non-significant pairwise correlations are indicated with an 'X'.

Discussion

Strain-specific estimates of host tissue abundance predict lifetime host-totick transmission

Our primary measure of host-to-tick transmission (HTT) for the 11 strains of *B. burgdorferi* was the nymphal infection prevalence (NIP), a common metric of HTT of *B. burgdorferi* sl genospecies [36,37,56]. We found that our estimates of lifetime NIP (hereafter lifetime HTT) for the 11 strains of *B. burgdorferi* were significantly positively correlated with the spirochete load in host tissues, such as the heart, bladder, and ear biopsies (**Figs 3 and 4**). Our results agree with previous studies that found a positive phenotypic correlation between host tissue spirochete load and HTT [13,26,27]. The pairwise correlation matrix found that all four transmission measures, LIP, LSL, NIP, and NSL, were strongly correlated with each other and they were strongly correlated with the spirochete loads in the mouse heart, bladder, and ear biopsies (**Fig 5**). The demonstration of a positive correlation between host tissue spirochete load and



Fig 6. Our laboratory-based estimates of the lifetime host-to-tick transmission for the 11 strains of *B. burgdorferi* **are positively related to the frequencies of these strains in natural populations of** *I. scapularis* **ticks in North America.** The strain-specific frequencies were combined for both the northeastern (blue) and midwestern (red) strains and did not adjust for differential sampling effort between regions. The slope of the logistic regression is positive and significant (slope = 9.685, SE = 3.724, p = 0.029). The error bars represent the 95% confidence intervals.

HTT across 11 genetically distinct strains of *B. burgdorferi* suggests an underlying genetic correlation between these two life history traits, which shapes their evolution. The present study suggests that strains of *B. burgdorferi* are under strong selection to maintain high as possible abundance in the tissues of their reservoir hosts to achieve high HTT to feeding *I. scapularis* ticks. In summary, this study demonstrates the following causal chain linking spirochete replication inside the host to lifetime HTT, high spirochete load in the host tissues increases the probability that feeding larvae will acquire spirochetes, and these engorged infected larvae have a high probability of moulting into infected nymphs.

Pathogen life history theory typically assumes that the cost of maintaining high pathogen abundance in the host tissues is reduced host survival (*i.e.*, increased virulence) [1–6]. Thus, higher virulence typically acts as a brake on higher pathogen abundance in the host tissues [1–6]. However, there is little evidence that infection with *B. burgdorferi* influences the survival of adult *P. leucopus* [38–40], although there is some evidence that infection with *B. afzelii* influences the reproduction of bank voles (*Myodes glareolus*) [57]. Infected *P. leucopus* mice develop a strong antibody response against *B. burgdorferi* that likely helps to control spirochete loads in the mouse tissues [25,58]. Infection with *B. burgdorferi* does not induce obvious pathology in the tissues of adult *P. leucopus* mice [41,59], although one study found pathology in juvenile *P. leucopus* [41]. A review of the reservoir competence of *Peromyscus* mice emphasized that their immune system has evolved to be tolerant of *B. burgdorferi* sl and other tickborne pathogens [40]. It is likely that there is strong selection against virulence of

B. burgdorferi in its rodent reservoir hosts as persistence of *B. burgdorferi* transmission demands mobility of the hosts to contact the tick vector which, unlike flying hematophagous insects, has limited mobility and adopts an ambush strategy [60].

Host tissue type and transmission of B. burgdorferi to feeding ticks

Many vector-borne pathogens maintain high abundance in the blood to achieve transmission to feeding arthropod vectors [12,61,62]. Borrelia burgdorferi has low abundance in the blood of vertebrate hosts [63] and bacteremia is not important for host-to-tick transmission [64]. For B. burgdorferi sl pathogens, the host skin is believed to be the critical organ for transmission to feeding *Ixodes* ticks, and this has been demonstrated experimentally for *B. afzelii* [13,26,27]. It could be argued that the spirochete loads in the ear biopsies would be expected to have a stronger relationship with the NIP compared to the spirochete loads in the tissue samples collected at necropsy. The reason is because the ear biopsies were sampled the day before each of the three larval infestations, and therefore provide a snapshot of the mouse tissue spirochete load immediately prior to each of the three transmission events. For the mice in this study, we had previously shown that the ear biopsy spirochete loads remained constant over time for some strains, whereas they decreased dramatically over time for other strains [42]. As expected, the abundance of *B. burgdorferi* in the ear biopsies sampled the day before each of the three larval infestations was a good predictor of the NIP (Fig 4). This result agrees with previous studies on *B. afzelii* that have found positive relationships between the spirochete load in ear biopsies and the infection prevalence in feeding Ixodes ticks [13,27]. One reason why ear biopsies may be less reliable than necropsy tissues is because they are a much smaller sample of skin and because repeated biopsy sampling damages the ear tissue, which probably has a negative impact on the resident spirochete population.

Given that the skin is believed to be the critical organ for host-to-tick transmission [13,26,27], it could be considered surprising that the spirochete loads in the internal organs (e.g., heart and bladder) at necropsy were better predictors of the NIP compared to the spirochete loads in the skin-related tissues (e.g., ventral skin, left ear, right ear). However, in infected hosts there is possibly a complex dynamic of movement of spirochaetes amongst internal organs and the skin from where they are transmitted, involving chemotaxis of spirochaetes from internal organs to the skin, and from skin to tick, associated with responses to tick saliva [65,66], with depletion of spirochaete loads in skin where ticks are feeding (particularly the ears from where mice have difficulty grooming ticks off). The hypothesis that the larval infestation decreased the spirochete load in the skin explains the following three results. First, the infection prevalence and the spirochete loads of the skin-related tissues (ventral skin, left ear, right ear) at necropsy (*i.e.*, shortly after the third larval infestation) were much lower compared to the internal organs (heart, bladder). Second, infection prevalence was 91.7% in the third biopsy of the right ear on day 89 PI (prior to tick feeding) and then dropped to 14.3% in the right ear sample collected at necropsy on day 97 PI, which was 7 days after the third larval tick infestation. Third, the mean spirochete loads of the three biopsies of the right ears were correlated with the spirochete loads in the heart and bladder collected at necropsy, but not with the spirochete loads in any of the skin samples, including the right ear, collected at necropsy (Fig 5). In summary, we believe that the larval infestation on day 90 PI transiently reduced the spirochete loads in the skin-related tissues, but not in the internal organs like the heart and bladder. As a result, the heart and bladder were better predictors of the lifetime NIP compared to the skin-related tissues. Future studies should investigate whether larval infestation influences the spirochete load in different host tissues.

Strain-specific estimates of lifetime HTT are correlated with strain-specific frequencies in nature

Our strain-specific estimates of lifetime HTT were significantly positively correlated with the frequencies of these strains in field and surveillance studies (**Fig 6**). The strains that have higher HTT in an experimental setting were also the ones that are more commonly found in *I. scapularis* ticks in nature in Canada and the USA. We recently showed that the mean spirochete load for the subset of infected mouse tissues was significantly positively correlated with strain frequency [42]. In that study, we calculated the mean spirochete loads without standardizing each mouse tissue to z-scores and this mean was therefore dominated by the spirochete loads in the heart and bladder (i.e., because these tissues have high infection prevalence and high spirochete loads) [42]. The positive relationship between lifetime HTT and strain-specific frequency in the present study agrees with a previous study on *B. afzelii* that found that laboratory-based estimates of the R₀ values of six *ospC* strains were related to their frequencies at a local field site in Neuchatel, Switzerland [18]. The present study is more general because the strain-specific frequencies were obtained from *I. scapularis* tick populations that were distributed across North America [46,53].

The estimation of unbiased MLST frequencies in nature using the published literature is complicated. The ticks in the study by Ogden et al [46] were collected from humans and companion animals as part of national passive surveillance system in Canada. Passive surveillance may result in skewed estimates of pathogen strain structure because ticks are mostly collected from humans and domestic pets, rather than from the environment or from natural hosts, and because passive surveillance also collects adventitious ticks that may have been dispersed long distances from their origin by migratory birds [67]. The ticks in the study by Travinsky et al [53] had been collected using a standardized sampling protocol at 40 different field sites in the northeastern, mid-Atlantic, and north central (midwestern) USA [53,68,69]. Both studies indicated whether the MLSTs were from the northeastern or the midwestern (north central) USA, but some MLSTs occurred in multiple geographic regions. The study by Ogden et al [46] had 15.6 times more MLSTs from the Northeast compared to the Midwest. For this reason, we calculated the MLST frequencies for the two geographic regions combined (frequency 2) and separately (frequency 4). Other complications included strains that had no matching MLST in the published data sets (Strains Bb16-66 and Bb16-54) and strains that matched with respect to their MLST but not their *ospC* type (Bb16-10-2 and Bb16-57-1). Despite these challenges, we found a positive relationship between our laboratory estimates of lifetime HTT and frequency in I. scapularis ticks in North America across 11 strains of B. burgdorferi. This result suggests that strains that have the highest fitness in M. musculus mice in the laboratory (even though *M. musculus* is not a natural host of *B. burgdorferi*) are those that are most common in nature, and that laboratory studies can enhance our understanding of pathogen strain structure in nature.

Previous studies have shown that strains of *B. burgdorferi* can be transmitted by many different vertebrate hosts including mice, chipmunks, squirrels, shrews, and other species [30, 70–73]. However, studies have identified considerable variation among vertebrate host species in their contribution to the epidemiology of *B. burgdorferi* [56,73,74], which depends on a number of factors including (1) host density, (2) numbers of larval ticks that are fed by the host, (3) probability that the host is infected with *B. burgdorferi*, and (4) probability that ticks acquire *B. burgdorferi* after feeding on an infected host (often referred to as host reservoir competence or host infectivity) [56,73,74]. Numerous studies have suggested that rodents, such as *P. leucopus* and the eastern chipmunk (*Tamias striatus*), are critical for determining the density of infected nymphs (DIN) and the nymphal infection prevalence (NIP) in an area [75–77]. Rodent-targeted interventions that are highly effective at reducing the DIN and/or NIP provide further evidence that *P. leucopus* is a keystone species for the epidemiology of *B. burgdorferi*. Thus, *B. burgdorferi* strains that are specialized for host species that are abundant and have high reservoir competence, such as *P. leucopus*, will have high frequency in nature. Strains that are specialized for host species that are rare and/or have low reservoir competence will have low frequency in nature.

The multiple niche polymorphism (MNP) hypothesis suggests that different B. burgdorferi strains are adapted to different host species [30,70-72,78]. Evidence for the MNP hypothesis comes from field studies showing that different B. burgdorferi strains are associated with different host species [30,70-72]. Some studies on B. burgdorferi in North America have found hoststrain associations [30,71], whereas others have not [70,72]. One criticism of all these studies [30,70–72] is that they do not control for spatiotemporal variation in exposure. Another criticism of these studies is their small sample sizes (n < 100 animals). The most comprehensive study on host-strain associations was done on 7 B. afzelii ospC strains in 3 different host species (bank vole, yellow-necked mouse, common shrew) with a sample size of ~2800 animals [19]. This study found no evidence of host-strain associations; common strains had high frequencies in all three host species and vice versa for rare strains [19]. The bank vole (*M. glareolus*) and yellow-necked mouse (Apodemus flavicollis) belong to the Cricetidae and the Muridae, respectively, which shared a common ancestor 35 million years ago (mya). Rodents and shrews belong to Euarchontoglires and Laurasiatheria, respectively, which shared a common ancestor ~100 mya [79]. The observation that the frequencies of these B. afzelii ospC strains is highly correlated across these three divergent host species suggests that the role of host-strain interactions is negligible for B. afzelii. With respect to our study, P. leucopus and M. musculus belong to the Cricetidae and the Muridae, respectively, which shared a common ancestor 35 mya [40]. An experimental infection study that used both these host species and two strains of *B. burg*dorferi found a significant host-strain interaction [34]. However, the pattern of HTT was generally the same in the two host species with no changes in the ranking of the strains between hosts [34]. We suggest that the strains that have high HTT in our C3H/HeJ lab mice are also the strains that have high HTT in common rodent reservoirs such as P. leucopus. It is possible that strains that have short-lived transmissible infections and/or relatively low HTT may be better adapted to species other than *P. leucopus* mice, such as voles, chipmunks, shrews, or birds. Future experimental infection studies should investigate whether fitness of different B. burgdorferi strains is correlated across different vertebrate hosts.

Mixed strain infections in the host and interactions among *B. burgdorferi* strains

In nature, vertebrate reservoir hosts are often infected with multiple strains of the same *B. burgdorferi* sl pathogen. Field studies on *B. burgdorferi* in North America [30,80] and on *B. afzelii* in Europe [81–83] have shown that mixed strain infections are the norm in endemic areas. Mixed strain infections in the vertebrate host result in competitive interactions that reduce host-to-tick transmission of strains to feeding *Ixodes* ticks and this phenomenon has been shown for *B. burgdorferi* in *P. leucopus* [31,36] and for *B. afzelii* in *M. musculus* [26,32,33]. The mechanism underlying competition between strains of *B. burgdorferi* sl is not known, but strain abundance in host tissues may be important. An experimental infection study of BALB/c mice with two strains of *B. afzelii* found that the strain that established the higher spirochete load in the tissues of co-infected mice had higher HTT to feeding *I. ricinus* ticks [26]. Thus, high abundance in the host tissues, whether alone or in mixed infections, enhances strain transmission to feeding ticks. Studies on the rodent malaria parasite,

Plasmodium chabaudi, have shown that strains that establish the highest abundance in the blood when alone in the rodent host are also the strains that are the most competitive in mixed strain infections [61]. Whether this phenomenon is true for *B. burgdorferi* remains to be tested.

Male mice have higher transmission of B. burgdorferi to I. scapularis ticks

An interesting result was that the lifetime HTT was higher for male mice compared to female mice and this was true for 9 of 11 strains of *B. burgdorferi* (Fig 2). We had recently shown that male mice have significantly higher spirochete loads in their tissues, especially the skin, compared to female mice [42,84]. For the C3H/HeJ mice in the present study, the spirochete load in the ventral skin of the male mice was 15.3 times higher compared to female mice [42]. In another study on C3H/HeJ mice where B. burgdorferi was re-activated by the application of immunosuppressive corticosteroids, the spirochete load in the skin (averaged over four different sites) of the male mice was 72.6 times higher compared to female mice [84]. Testosteronebased immunosuppression is a general explanation why males often have higher pathogen abundance in their tissues than females [85–87]. The present study suggests that male mice have higher HTT because they have higher spirochete loads in their tissues compared to female mice. Field studies on P. leucopus in the USA and on bank voles (M. glareolus) in Sweden have shown that male rodents are more likely to be infected with *B. burgdorferi* sl than their female counterparts [38,88–90]. One reason for this sex bias in infection prevalence in the field is that male rodents have larger home ranges than female rodents and are more likely to encounter B. burgdorferi-infected nymphs [38,91–94]. For the same reasons, male rodents are more likely to encounter and feed a greater number of larval ticks [38,91-94]. Assuming similar survival rates between the sexes, our study suggests that male rodents contribute more to the prevalence of infected ticks in nature and hence to the risk of Lyme disease compared to their female counterparts. Future studies should investigate whether similar sex-specific differences in tissue spirochete load and HTT occur in wild rodent reservoir hosts.

HTT decreases over time for most B. burgdorferi strains

Interestingly, 10 of 11 *B. burgdorferi* strains had HTT greater than 90% for the first larval infestation (Fig 1), suggesting that there is little variation in fitness among these strains early in the infection. Pathogen life history theory predicts that early transmission is more important for pathogen fitness than late transmission, and this is especially true if the hosts are small, shortlived prey species. Field studies suggest that most *P. leucopus* mice do not survive to see a second Lyme disease transmission season [38,40,58,64,95]. Capture-mark-recapture studies in the field suggest that monthly survival rates of adult *P. leucopus* can be as low as 50% [38]. Assuming a host mortality rate of 50%, a strain that reaches peak HTT at 30 days PI will be twice as fit compared to a strain that reaches peak HTT at 60 days PI. Thus, all strains are under strong selection to have high HTT over the early phase of the infection (i.e., the first 30 days). Nevertheless, the observation that all strains established persistent infection in the mice suggests that this trait must have adaptive value. The fact that *B. burgdorferi* has evolved complicated antigenic variation mechanisms that allow it to evade the host antibody response [96] provides further evidence that persistent infection in the host must provide some fitness benefits.

Lifetime HTT remained above 90% over the course of the infection for 3 strains from the northeast (Bb16-174, Bb16-178-1, and Bb16-198 in Fig 1). For the other 8 strains, HTT decreased over the three successive infestations, which agrees with other studies [34,35,97,98], and the magnitude of the decrease determined variation in lifetime HTT (Fig 1). Interestingly,

the 3 strains with the most dramatic decline in HTT over time (Bb16-22-2, Bb16-54, and Bb16-66 in Fig 1) are from the Midwest. Differences in climate between the Northeast and Midwest cause differences in the seasonal phenology of immature I. scapularis ticks, which might explain these differences in lifetime HTT. In North America, nymphs become active and infect the reservoir hosts in the spring and early summer; the larvae subsequently acquire the spirochetes from these infected hosts in the late summer and early fall [23,64]. In the Northeast, the time interval between the larval peak and the nymphal peak is longer and their questing activity is therefore less synchronous compared to the Midwest [99]. Thus, strains in the Northeast must persist for longer in their rodent reservoir hosts compared to strains in the Midwest. Taken together, these observations suggest that the more synchronous questing behaviour of immature I. scapularis ticks in the Midwest relaxes selection on midwestern strains to have persistently high HTT over the course of the infection. There were too few strains studied here to explore this, but indeed the three strains that had constantly high HTT at all infestations were from the northeast (Bb16-174, Bb16-178-1 and Bb16-198 from Nova Scotia), while the three strains with the lowest HTT at the second and third infestations were from the Midwest (Bb16-22-2, Bb16-54 and Bb16-66 from Manitoba).

Conclusions

We tested the variation in host-to-tick transmission among strains of *B. burgdorferi* using a laboratory model host. We found that these estimates of HTT were positively correlated with measures of host tissue spirochete load, and with the strain frequencies in nature. From this we suggest a simple mechanistic explanation, where strains that establish a higher spirochete load in the tissues of the vertebrate host are more likely to be acquired by feeding larvae, which in turn results in more infected questing nymphs. The strains with higher spirochete loads in the host tissues and higher lifetime host-to-tick transmission are thus more common in nature.

Supporting information

S1 Information. Electronic supplementary material (ESM). (PDF)

Acknowledgments

The authors would like to thank Dr Susan Detmer and Dr Pini Zvionow for their assistance in necropsy and tissue collection. We would like to thank Georgia Hurry, Alexandra Foley-Eby and the staff of the WCVM animal care unit for their assistance with the experimental infection study. We would like to thank Dr Robbin Lindsay and Antonia DiBernardo for kindly providing the strains of *B. burgdorferi* that were used in this study, and for their advice on how to culture these strains. We thank Samir Mechai for re-checking the *ospC* types of the whole genome sequences of the PHAC strains of *B. burgdorferi*.

Author Contributions

Conceptualization: Christopher B. Zinck, Dustin Brisson, Nicholas H. Ogden, Maarten J. Voordouw.

Data curation: Christopher B. Zinck, Eva-Maria E. Uhlemann.

Formal analysis: Christopher B. Zinck, Catherine Bouchard, Maarten J. Voordouw.

Funding acquisition: Catherine Bouchard, Nicholas H. Ogden, Maarten J. Voordouw.

- Investigation: Christopher B. Zinck, Prasobh Raveendram Thampy, Eva-Maria E. Uhlemann, Hesham Adam, Jenny Wachter, Danae Suchan, Andrew D. S. Cameron, Ryan O. M. Rego.
- Methodology: Christopher B. Zinck, Prasobh Raveendram Thampy, Eva-Maria E. Uhlemann, Jenny Wachter, Danae Suchan, Andrew D. S. Cameron, Ryan O. M. Rego.
- Project administration: Christopher B. Zinck, Maarten J. Voordouw.
- Resources: Maarten J. Voordouw.
- Supervision: Christopher B. Zinck, Maarten J. Voordouw.
- Validation: Christopher B. Zinck, Prasobh Raveendram Thampy, Eva-Maria E. Uhlemann, Hesham Adam, Jenny Wachter, Danae Suchan, Andrew D. S. Cameron, Ryan O. M. Rego.
- Visualization: Christopher B. Zinck, Maarten J. Voordouw.
- Writing original draft: Christopher B. Zinck, Maarten J. Voordouw.
- Writing review & editing: Christopher B. Zinck, Jenny Wachter, Dustin Brisson, Catherine Bouchard, Maarten J. Voordouw.

References

- Alizon S, Hurford A, Mideo N, Van Baalen M. Virulence evolution and the trade-off hypothesis: history, current state of affairs and the future. J Evol Biol. 2009; 22(2):245–59. https://doi.org/10.1111/j.1420-9101.2008.01658.x WOS:000262516800002. PMID: 19196383
- Mackinnon MJ, Gandon S, Read AF. Virulence evolution in response to vaccination: The case of malaria. Vaccine. 2008; 26:C42–C52. https://doi.org/10.1016/j.vaccine.2008.04.012 WOS:000258164200007. PMID: 18773536
- Frank SA. Models of parasite virulence. Q Rev Biol. 1996; 71(1):37–78. ISI:A1996UE27500002. https://doi.org/10.1086/419267 PMID: 8919665
- Cressler CE, McLeod DV, Rozins C, van den Hoogen J, Day T. The adaptive evolution of virulence: a review of theoretical predictions and empirical tests. Parasitology. 2016; 143(7):915–30. https://doi.org/ 10.1017/S003118201500092X WOS:000376781100011. PMID: 26302775
- Anderson RM, May RM. Coevolution of hosts and parasites. Parasitology. 1982; 85(OCT):411–26. https://doi.org/10.1017/s0031182000055360 WOS:A1982PP03200017. PMID: 6755367
- Visher E, Evensen C, Guth S, Lai E, Norfolk M, Rozins C, et al. The three Ts of virulence evolution during zoonotic emergence. P Roy Soc B-Biol Sci. 2021;288(1956). <u>https://doi.org/10.1098/rspb.2021</u>. 0900 WOS:000685237300005. PMID: 34375554
- Fraser C, Hollingsworth TD, Chapman R, de Wolf F, Hanage WP. Variation in HIV-1 set-point viral load: Epidemiological analysis and an evolutionary hypothesis. Proc Natl Acad Sci U S A. 2007; 104 (44):17441–6. https://doi.org/10.1073/pnas.0708559104 WOS:000250638400039. PMID: 17954909
- Mackinnon MJ, Read AF. Virulence in malaria: an evolutionary viewpoint. Philos T Roy Soc B. 2004; 359(1446):965–86. https://doi.org/10.1098/rstb.2003.1414 WOS:000222444800008. PMID: 15306410
- de Roode JC, Yates AJ, Altizer S. Virulence-transmission trade-offs and population divergence in virulence in a naturally occuring butterfly parasite. Proc Natl Acad Sci U S A. 2008; 105(21):7489–94. https://doi.org/10.1073/pnas.0710909105 WOS:000256378100029. PMID: 18492806
- Jensen KH, Little T, Skorping A, Ebert D. Empirical support for optimal virulence in a castrating parasite. PLOS Biol. 2006; 4(7):1265–9. https://doi.org/10.1371/journal.pbio.0040197 WOS:000238974300018. PMID: 16719563
- 11. Stearns SC. The evolution of life histories. Oxford: Oxford University Press; 1992.
- Mackinnon MJ, Read AF. Genetic relationships between parasite virulence and transmission in the rodent malaria *Plasmodium chabaudi*. Evolution. 1999; 53(3):689–703. <u>https://doi.org/10.2307/</u> 2640710 WOS:000081507300004.
- **13.** Raberg L. Infection intensity and infectivity of the tick-borne pathogen *Borrelia afzelii*. J Evol Biol. 2012; 25:1448–53.
- 14. Schneider P, Bell AS, Sim DG, O'Donnell AJ, Blanford S, Paaijmans KP, et al. Virulence, drug sensitivity and transmission success in the rodent malaria, *Plasmodium chabaudi*. P Roy Soc B-Biol Sci. 2012;

279(1747):4677-85. https://doi.org/10.1098/rspb.2012.1792 WOS:000310218400021. PMID: 23015626

- Acevedo MA, Dillemuth FP, Flick AJ, Faldyn MJ, Elderd BD. Virulence-driven trade-offs in disease transmission: A meta-analysis. Evolution. 2019; 73(4):636–47. https://doi.org/10.1111/evo.13692 WOS:000467993300001. PMID: 30734920
- Bambini S, Piet J, Muzzi A, Keijzers W, Comandi S, De Tora L, et al. An analysis of the sequence variability of meningococcal fHbp, NadA and NHBA over a 50-year period in the Netherlands. PLOS ONE. 2013; 8(5). https://doi.org/10.1371/journal.pone.0065043 WOS:000319385300005. PMID: 23717687
- Buckee CO, Gupta S, Kriz P, Maiden MCJ, Jolley KA. Long-term evolution of antigen repertoires among carried meningococci. P Roy Soc B-Biol Sci. 2010; 277(1688):1635–41. https://doi.org/10.1098/ rspb.2009.2033 WOS:000276997700003. PMID: 20129981
- Durand J, Jacquet M, Rais O, Gern L, Voordouw MJ. Fitness estimates from experimental infections predict the long-term strain structure of a vector-borne pathogen in the field. Scientific Reports. 2017; 7 (1851). https://doi.org/10.1038/s41598-017-01821-1 PMID: 28500292
- Raberg L, Hagstrom A, Andersson M, Bartkova S, Scherman K, Strandh M, et al. Evolution of antigenic diversity in the tick-transmitted bacterium *Borrelia afzelii*: a role for host specialization? J Evol Biol. 2017; 30(5):1034–41. https://doi.org/10.1111/jeb.13075 WOS:000400783800015. PMID: 28345277
- Weinberger DM, Trzcinski K, Lu Y-J, Bogaert D, Brandes A, Galagan J, et al. Pneumococcal capsular polysaccharide structure predicts serotype prevalence. PLOS Pathog. 2009; 5(6). https://doi.org/10. 1371/journal.ppat.1000476 WOS:000268444500023. PMID: 19521509
- Steere AC, Strle F, Wormser GP, Hu LDT, Branda JA, Hovius JR, et al. Lyme borreliosis. Nature Reviews Disease Primers. 2016;2. https://doi.org/10.1038/nrdp.2016.90 WOS:000397870500001. PMID: 27976670
- 22. Stanek G, Reiter M. The expanding Lyme *Borrelia* complex-clinical significance of genomic species? Clin Microbiol Infec. 2011; 17(4):487–93. https://doi.org/10.1111/j.1469-0691.2011.03492.x WOS:000288501700004. PMID: 21414082
- Kurtenbach K, Hanincova K, Tsao JI, Margos G, Fish D, Ogden NH. Fundamental processes in the evolutionary ecology of Lyme borreliosis. Nat Rev Microbiol. 2006; 4:660–9. https://doi.org/10.1038/ nrmicro1475 PMID: 16894341
- Piesman J, Gern L. Lyme borreliosis in Europe and North America. Parasitology. 2004; 129:S191– S220. https://doi.org/10.1017/s0031182003004694 WOS:000230134600013. PMID: 15938512
- Baum E, Hue F, Barbour AG. Experimental infections of the reservoir species *Peromyscus leucopus* with diverse strains of *Borrelia burgdorferi*, a Lyme disease agent. mBio. 2012; 3(6):e00434–12. <u>https:// doi.org/10.1128/mBio.00434-12 MEDLINE:PMID: 23221801.</u>
- Genné D, Rossel M, Sarr A, Battilotti F, Rais O, Rego ROM, et al. Competition between strains of *Borrelia afzelii* in the host tissues and consequences for transmission to ticks. The ISME Journal. 2021. https://doi.org/10.1038/s41396-021-00939-5 PMID: 33658621
- Jacquet M, Durand J, Rais O, Voordouw MJ. Cross-reactive acquired immunity influences transmission success of the Lyme disease pathogen, *Borrelia afzelii*. Infection Genetics and Evolution. 2015; 36:131–40. https://doi.org/10.1016/j.meegid.2015.09.012 WOS:000367548300017. PMID: 26384476
- Wang G, Ojaimi C, Iyer R, Saksenberg V, McClain SA, Wormser GP, et al. Impact of genotypic variation of *Borrelia burgdorferi* sensu stricto on kinetics of dissemination and severity of disease in C3H/HeJ mice. Infect Immun. 2001; 69(7):4303–12.
- Wang GQ, Ojaimi C, Wu HY, Saksenberg V, Iyer R, Liveris D, et al. Disease severity in a murine model of Lyme borreliosis is associated with the genotype of the infecting *Borrelia burgdorferi* sensu stricto strain. J Infect Dis. 2002; 186(6):782–91. <u>https://doi.org/10.1086/343043</u> WOS:000177694800008. PMID: 12198612
- Brisson D, Dykhuizen DE. ospC diversity in Borrelia burgdorferi: different hosts are different niches. Genetics. 2004; 168:713–22.
- Derdakova M, Dudioak V, Brei B, Brownstein JS, Schwartz I, Fish D. Interaction and transmission of two Borrelia burgdorferi sensu stricto strains in a tick-rodent maintenance system. Appl Environ Microbiol. 2004; 70(11):6783–8. https://doi.org/10.1128/aem.70.11.6783–6788.2004 ISI:000225076100054.
- **32.** Genné D, Sarr A, Gomez-Chamorro A, Durand J, Cayol C, Rais O, et al. Competition between strains of *Borrelia afzelii* inside the rodent host and the tick vector. P Roy Soc B-Biol Sci. 2018; 285(1890). https://doi.org/10.1098/rspb.2018.1804 PMID: 30381382
- Genné D, Sarr A, Rais O, Voordouw MJ. Competition between strains of *Borrelia afzelii* in immature *Ixodes ricinus* ticks is not affected by season. Front Cell Infect Microbiol. 2019; 9(431):1–14. https://doi. org/10.3389/fcimb.2019.00431 PMID: 31921706

- Hanincova K, Ogden NH, Diuk-Wasser M, Pappas CJ, Iyer R, Fish D, et al. Fitness variation of *Borrelia burgdorferi* sensu stricto strains in mice. Appl Environ Microbiol. 2008; 74(1):153–7. https://doi.org/10. 1128/aem.01567-07 ISI:000252195200018. PMID: 17981941
- Jacquet M, Margos G, Fingerle V, Voordouw MJ. Comparison of the lifetime host-to-tick transmission between two strains of the Lyme disease pathogen *Borrelia afzelii*. Parasit Vectors. 2016;9. https://doi. org/10.1186/s13071-016-1929-z WOS:000392371300002. PMID: 27986081
- Rynkiewicz EC, Brown J, Tufts DM, Huang C-I, Kampen H, Bent SJ, et al. Closely-related *Borrelia burg-dorferi* (sensu stricto) strains exhibit similar fitness in single infections and asymmetric competition in multiple infections. Parasit Vectors. 2017; 10(64):1–9. https://doi.org/10.1186/s13071-016-1964-9 PMID: 28166814
- Tonetti N, Voordouw MJ, Durand J, Monnier S, Gern L. Genetic variation in transmission success of the Lyme borreliosis pathogen *Borrelia afzelii*. Ticks Tick Borne Dis. 2015; 6(3):334–43. https://doi.org/10. 1016/j.ttbdis.2015.02.007 WOS:000354158100021. PMID: 25748511
- Voordouw MJ, Lachish S, Dolan MC. The Lyme disease pathogen has no effect on the survival of its rodent reservoir host. PLOS ONE. 2015; 10(2). https://doi.org/10.1371/journal.pone.0118265 WOS:000350322700110. PMID: 25688863
- Schwanz LE, Voordouw MJ, Brisson D, Ostfeld RS. *Borrelia burgdorferi* has minimal impact on the Lyme disease reservoir host *Peromyscus leucopus*. Vector-Borne Zoonot. 2011; 11(2):117–24. https:// doi.org/10.1089/vbz.2009.0215 WOS:000286871900003. PMID: 20569016
- 40. Barbour AG. Infection resistance and tolerance in *Peromyscus* spp., natural reservoirs of microbes that are virulent for humans. Semin Cell Dev Biol. 2017; 61:115–22. https://doi.org/10.1016/j.semcdb.2016. 07.002 WOS:000391837800015. PMID: 27381345
- 41. Moody KD, Terwilliger GA, Hansen GM, Barthold SW. Experimental *Borrelia burgdorferi* infection in *Peromyscus leucopus*. J Wildl Dis. 1994; 30(2):155–61.
- 42. Zinck CB, Thampy PR, Rego ROM, Brisson D, Ogden NH, Voordouw M. Borrelia burgdorferi strain and host sex influence pathogen prevalence and abundance in the tissues of a laboratory rodent host. Mol Ecol. 2022;n/a(n/a). https://doi.org/10.1111/mec.16694 PMID: 36112076
- 43. Tyler S, Tyson S, Dibernardo A, Drebot M, Feil EJ, Graham M, et al. Whole genome sequencing and phylogenetic analysis of strains of the agent of Lyme disease *Borrelia burgdorferi* from Canadian emergence zones. Scientific Reports. 2018;8. <u>https://doi.org/10.1038/s41598-018-28908-7</u> WOS:000438343600062. PMID: 30002414
- 44. Margos G, Tsao JI, Castillo-Ramirez S, Girard YA, Hamer SA, Hoen AG, et al. Two boundaries separate *Borrelia burgdorferi* populations in North America. Appl Environ Microbiol. 2012; 78(17):6059–67. https://doi.org/10.1128/aem.00231-12 WOS:000307611800010. PMID: 22729536
- Mechai S, Margos G, Feil EJ, Lindsay LR, Ogden NH. Complex population structure of *Borrelia burgdor-feri* in southeastern and south central Canada as revealed by phylogeographic analysis. Appl Environ Microbiol. 2015; 81(4):1309–18. <u>https://doi.org/10.1128/aem.03730-14</u> WOS:000348927600013. PMID: 25501480
- 46. Ogden NH, Margos G, Aanensen DM, Drebot MA, Feil EJ, Hanincova K, et al. Investigation of genotypes of *Borrelia burgdorferi* in *Ixodes scapularis* ticks collected during surveillance in Canada. Appl Environ Microbiol. 2011; 77(10):3244–54. https://doi.org/10.1128/aem.02636-10 WOS:000290473200011. PMID: 21421790
- Dolan MC, Piesman J, Schneider BS, Schriefer M, Brandt K, Zeidner NS. Comparison of disseminated and nondisseminated strains of *Borrelia burgdorferi* sensu stricto in mice naturally infected by tick bite. Infect Immun. 2004; 72(9):5262–6. <u>https://doi.org/10.1128/iai.72.9.5262–5266.2004</u> WOS:000223580400038.
- Ma Y, Seiler KP, Eichwald EJ, Weis JH, Teuscher C, Weis JJ. Distinct characteristics of resistance to Borrelia burgdorferi induced arthritis in C57BL/6N mice. Infect Immun. 1998; 66(1):161–8. WOS:00007129000023.
- Barthold SW, Persing DH, Armstrong AL, Peeples RA. Kinetics of *Borrelia burgdorferi* dissemination and evolution of disease after intradermal inoculation of mice. Am J Pathol. 1991; 139(2):263–73. WOS:A1991GA98200004.
- Courtney JW, Kostelnik LM, Zeidner NS, Massung RF. Multiplex real-time PCR for detection of Anaplasma phagocytophilum and Borrelia burgdorferi. J Clin Microbiol. 2004; 42(7):3164–8. https://doi.org/ 10.1128/jcm.42.7.3164–3168.2004 WOS:000222672100046.
- Schrader C, Schielke A, Ellerbroek L, Johne R. PCR inhibitors—occurrence, properties and removal. J Appl Microbiol. 2012; 113(5):1014–26. https://doi.org/10.1111/j.1365-2672.2012.05384.x WOS:000310281100002. PMID: 22747964
- 52. Sidstedt M, Hedman J, Romsos EL, Waitara L, Wadso L, Steffen CR, et al. Inhibition mechanisms of hemoglobin, immunoglobulin G, and whole blood in digital and real-time PCR. Anal Bioanal Chem.

2018; 410(10):2569-83. https://doi.org/10.1007/s00216-018-0931-z WOS:000427797800010. PMID: 29504082

- Travinsky B, Bunikis J, Barbour AG. Geographic differences in genetic locus linkages for *Borrelia burg-dorferi*. Emerg Infect Dis. 2010; 16(7):1147–50. https://doi.org/10.3201/eid1607.091452 WOS:000279522200019. PMID: 20587192
- 54. Zinck CB, Raveendran Thampy P, Uhlemann E-M, Adam H, Wachter J, Suchan D, et al. Variation among strains of *Borrelia burgdorferi* in host tissue abundance and lifetime transmission determine the population strain structure in nature. In: Dryad, editor. 2023.
- 55. R Development Core Team. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2021.
- LoGiudice K, Ostfeld RS, Schmidt KA, Keesing F. The ecology of infectious disease: Effects of host diversity and community composition on Lyme disease risk. Proc Natl Acad Sci U S A. 2003; 100 (2):567–71. https://doi.org/10.1073/pnas.0233733100 PMID: 12525705
- Cayol C, Giermek A, Gomez-Chamorro A, Hytönen J, Kallio ER, Mappes T, et al. *Borrelia afzelii* alters reproductive success in a rodent host. P Roy Soc B-Biol Sci. 2018; 285(1884). https://doi.org/10.1098/ rspb.2018.1056 PMID: 30068677
- Bunikis J, Tsao J, Luke CJ, Luna MG, Fish D, Barbour AG. Borrelia burgdorferi infection in a natural population of *Peromyscus leucopus* mice: a longitudinal study in an area where Lyme borreliosis is highly endemic. J Infect Dis. 2004; 189:1515–23.
- 59. Wright SD, Nielsen SW. Experimental infection of the white-footed mouse with *Borrelia burgdorferi*. Am J Vet Res. 1990; 51(12):1980–7. WOS:A1990EL69600016.
- Randolph SE. Ticks are not insects: consequences of contrasting vector biology for transmission potential. Parasitol Today. 1998; 14(5):186–92. https://doi.org/10.1016/s0169-4758(98)01224-1 PMID: 17040748
- de Roode JC, Pansini R, Cheesman SJ, Helinski MEH, Huijben S, Wargo AR, et al. Virulence and competitive ability in genetically diverse malaria infections. Proc Natl Acad Sci U S A. 2005; 102(21):7624– 8. https://doi.org/10.1073/pnas.0500078102 ISI:000229417500041. PMID: 15894623
- 62. Komar N, Langevin S, Hinten S, Nemeth N, Edwards E, Hettler D, et al. Experimental infection of north American birds with the New York 1999 strain of West Nile virus. Emerg Infect Dis. 2003; 9(3):311–22. https://doi.org/10.3201/eid0903.020628 WOS:000181507700005. PMID: 12643825
- Lee WY, Moriarty TJ, Wong CHY, Zhou H, Strieter RM, van Rooijen N, et al. An intravascular immune response to *Borrelia burgdorferi* involves Kupffer cells and iNKT cells. Nat Immunol. 2010; 11(4):295– U36. https://doi.org/10.1038/ni.1855 WOS:000275849500008. PMID: 20228796
- Tsao J. Reviewing molecular adaptations of Lyme borreliosis spirochetes in the context of reproductive fitness in natural transmission cycles. Vet Res (Paris). 2009; 40(2). <u>https://doi.org/10.1051/vetres/</u> 2009019 PMID: 19368764
- Shih CM, Chao LL, Yu CP. Chemotactic migration of the Lyme disease spirochete (*Borrelia burgdorferi*) to salivary gland extracts of vector ticks. Am J Trop Med Hyg. 2002; 66(5):616–21. WOS:000177501000031.
- 66. Van Gundy TJ, Ullmann AJ, Brandt KS, Gilmore RD. A transwell assay method to evaluate *Borrelia burgdorferi* sensu stricto migratory chemoattraction toward tick saliva proteins. Ticks Tick Borne Dis. 2021; 12(5). https://doi.org/10.1016/j.ttbdis.2021.101782 WOS:000679958400001. PMID: 34274573
- Ogden NH, Trudel L, Artsob H, Barker IK, Beauchamp G, Charron DF, et al. *Ixodes scapularis* ticks collected by passive surveillance in Canada: Analysis of geographic distribution and infection with Lyme Borreliosis agent *Borrelia burgdorferi*. J Med Entomol. 2006; 43(3):600–9. https://doi.org/10.1603/0022-2585(2006)43[600:istcbp]2.0.co;2. WOS:000242788500024.
- Barbour AG, Bunikis J, Travinsky B, Hoen AG, Diuk-Wasser MA, Fish D, et al. Niche partitioning of *Borrelia burgdorferi* and *Borrelia miyamotoi* in the same tick vector and mammalian reservoir species. Am J Trop Med Hyg. 2009; 81(6):1120–31. <u>https://doi.org/10.4269/ajtmh.2009.09–0208</u> WOS:000272709600031.
- 69. Diuk-Wasser MA, Gatewood AG, Cortinas MR, Yaremych-Hamer S, Tsao J, Kitron U, et al. Spatiotemporal patterns of host-seeking Ixodes scapularis nymphs (Acari: Iodidae) in the United States. J Med Entomol. 2006; 43(2):166–76. https://doi.org/10.1603/0022-2585(2006)043[0166:Spohis]2.0.Co;2 WOS:000236184600006.
- 70. Hanincova K, Kurtenbach K, Diuk-Wasser M, Brei B, Fish D. Epidemic spread of Lyme borreliosis, Northeastern United States. Emerg Infect Dis. 2006; 12(4):604–11. <u>https://doi.org/10.3201/eid1204.</u> 051016 PMID: 16704808

- Mechai S, Margos G, Feil EJ, Barairo N, Lindsay LR, Michel P, et al. Evidence for host-genotype associations of *Borrelia burgdorferi* sensu stricto. PLOS ONE. 2016; 11(2). <u>https://doi.org/10.1371/journal.pone.0149345</u> WOS:000371276100061. PMID: 26901761
- 72. Vuong HB, Canham CD, Fonseca DM, Brisson D, Morin PJ, Smouse PE, et al. Occurrence and transmission efficiencies of *Borrelia burgdorferi ospC* types in avian and mammalian wildlife. Infect Genet Evol. 2014; 27:594–600. <u>https://doi.org/10.1016/j.meegid.2013.12.011</u> WOS:000343951900068. PMID: 24382473
- Brunner J, LoGiudice K, Ostfeld R. Estimating reservoir competence of *Borrelia burgdorferi* hosts: prevalence and infectivity, sensitivity, and specificity. J Med Entomol. 2008; 45(1):139–47.
- Mather TN, Wilson ML, Moore SI, Ribeiro JMC, Spielman A. Comparing the relative potential of rodents as reservoirs of the Lyme disease spirochete (*Borrelia burgdorferi*). Am J Epidemiol. 1989; 130(1):143– 50.
- 75. Ostfeld RS, Canham CD, Oggenfuss K, Winchcombe RJ, Keesing F. Climate, deer, rodents, and acorns as determinants of variation in Lyme-disease risk. PLOS Biol. 2006; 4(6):1058–68. https://doi. org/10.1371/journal.pbio.0040145 PMID: 16669698
- 76. Ostfeld RS, Schauber EM, Canham CD, Keesing F, Jones CG, Wolff JO. Effects of acorn production and mouse abundance on abundance and Borrelia burgdorferi infection prevalence of nymphal *Ixodes scapularis* ticks. Vector-Borne Zoonot. 2001; 1(1):55–+. https://doi.org/10.1089/153036601750137688 WOS:000210777500006. PMID: 12653136
- 77. Jones CG, Ostfeld RS, Richard MP, Schauber EM, Wolff JO. Chain reactions linking acorns to gypsy moth outbreaks and Lyme disease risk. Science. 1998; 279(5353):1023–6. https://doi.org/10.1126/ science.279.5353.1023 WOS:000072006400045. PMID: 9461433
- Adams B, Walter KS, Diuk-Wasser MA. Host Specialisation, Immune Cross-Reaction and the Composition of Communities of Co-circulating Borrelia Strains. Bull Math Biol. 2021; 83(6). <u>https://doi.org/10.1007/s11538-021-00896-2</u> WOS:000646565000001. PMID: 33939028
- 79. Springer MS, Murphy WJ, Eizirik E, O'Brien SJ. Placental mammal diversification and the Cretaceous-Tertiary boundary. Proc Natl Acad Sci U S A. 2003; 100(3):1056–61. <u>https://doi.org/10.1073/pnas.</u> 0334222100 WOS:000180838100053. PMID: 12552136
- Swanson KI, Norris DE. Presence of multiple variants of *Borrelia burgdorferi* in the natural reservoir *Per-omyscus leucopus* throughout a transmission season. Vector-Borne Zoonot. 2008; 8(3):397–405. https://doi.org/10.1089/vbz.2007.0222 WOS:000257135900013. PMID: 18399776
- Andersson M, Scherman K, Raberg L. Multiple-strain infections of *Borrelia afzelii*: a role for within-host interactions in the maintenance of antigenic diversity? Am Nat. 2013; 181(4):545–54.
- Strandh M, Raberg L. Within-host competition between *Borrelia afzelii* ospC strains in wild hosts as revealed by massively parallel amplicon sequencing. Philos T Roy Soc B. 2015; 370(1675):1–8. https://doi.org/10.1098/rstb.2014.0293 MEDLINE:PMID: 26150659.
- Pérez D, Kneubühler Y, Rais O, Jouda F, Gern L. Borrelia afzelii ospC genotype diversity in *Ixodes ricinus* questing ticks and ticks from rodents in two Lyme borreliosis endemic areas: Contribution of cofeeding ticks. Ticks Tick Borne Dis. 2011; 2(3):137–42.
- Bourgeois B, Koloski C, Foley-Eby A, Zinck CB, Hurry G, Boulanger N, et al. Clobetasol increases the abundance of *Borrelia burgdorferi* in the skin 70 times more in male mice compared to female mice. Ticks Tick Borne Dis. 2022; 13(6):102058. https://doi.org/10.1016/j.ttbdis.2022.102058.
- 85. Klein SL, Flanagan KL. Sex differences in immune responses. Nat Rev Immunol. 2016; 16(10):626–38. https://doi.org/10.1038/nri.2016.90 WOS:000384551900010. PMID: 27546235
- 86. Trigunaite A, Dimo J, Jorgensen TN. Suppressive effects of androgens on the immune system. Cell Immunol. 2015; 294(2):87–94. <u>https://doi.org/10.1016/j.cellimm.2015.02.004</u> WOS:000351480800006. PMID: 25708485
- vom Steeg LG, Klein SL. SeXX Matters in Infectious Disease Pathogenesis. PLOS Pathog. 2016; 12 (2). https://doi.org/10.1371/journal.ppat.1005374 WOS:000378152900005. PMID: 26891052
- Tschirren B, Andersson M, Scherman K, Westerdahl H, Mittl PRE, Raberg L. Polymorphisms at the innate immune receptor TLR2 are associated with *Borrelia* infection in a wild rodent population. P Roy Soc B-Biol Sci. 2013; 280(1759):20130364. https://doi.org/2013036410.1098/rspb.2013.0364 WOS:000317482100024.
- Zawada SG, von Fricken ME, Weppelmann TA, Sikaroodi M, Gillevet PM. Optimization of tissue sampling for *Borrelia burgdorferi* in white-footed mice (*Peromyscus leucopus*). PLOS ONE. 2020; 15(1). https://doi.org/10.1371/journal.pone.0226798 WOS:000534599100019. PMID: 31978068
- 90. Hamer SA, Hickling GJ, Sidge JL, Walker ED, Tsao JI. Synchronous phenology of juvenile lxodes scapularis, vertebrate host relationships, and associated patterns of *Borrelia burgdorferi* ribotypes in the

midwestern United States. Ticks Tick Borne Dis. 2012; 3(2):65–74. <u>https://doi.org/10.1016/j.ttbdis.</u> 2011.11.004 WOS:000304570200001. PMID: 22297162

- **91.** Devevey G, Brisson D. The effect of spatial heterogeneity on the aggregation of ticks on white-footed mice. Parasitology. 2012; 139:915–25.
- 92. Ostfeld RS, Brisson D, Oggenfuss K, Devine J, Levy MZ, Keesing F. Effects of a zoonotic pathogen, Borrelia burgdorferi, on the behavior of a key reservoir host. Ecol Evol. 2018; 8(8):4074–83. Epub 2018/ 05/04. https://doi.org/10.1002/ece3.3961 PMID: 29721281; PubMed Central PMCID: PMC5916280.
- Perkins SE, Cattadori IM, Tagliapietra V, Rizzoli AP, Hudson PJ. Empirical evidence for key hosts in persistence of a tick-borne disease. Int J Parasitol. 2003; 33:909–17. https://doi.org/10.1016/s0020-7519(03)00128-0 PMID: 12906875
- 94. Bouchard C, Beauchamp G, Nguon S, Trudel L, Milord F, Lindsay LR, et al. Associations between Ixodes scapularis ticks and small mammal hosts in a newly endemic zone in southeastern Canada: Implications for Borrelia burgdorferi transmission. Ticks Tick Borne Dis. 2011; 2(4):183–90. https://doi. org/10.1016/j.ttbdis.2011.03.005 WOS:000299246200002. PMID: 22108010
- **95.** Hofmeister EK, Ellis BA, Glass GE, Childs JE. Longitudinal study of infection with *Borrelia burgdorferi* in a population of *Peromyscus leucopus* at a Lyme disease-enzootic site in Maryland. Am J Trop Med Hyg. 1999; 60(4):589–609.
- 96. Graves C, Ros VID, Stevenson B, Sniegowski P, Brisson D. Natural selection promotes antigenic evolvability. PLOS Pathog. 2013; 9(11):e1003766. <u>https://doi.org/10.1371/journal.ppat.1003766</u> PMID: 24244173
- 97. Donahue JG, Piesman J, Spielman A. Reservoir competence of white-footed mice for Lyme disease spirochetes. Am J Trop Med Hyg. 1987; 36:92–6. https://doi.org/10.4269/ajtmh.1987.36.92 PMID: 3812887
- 98. Richter D, Klug B, Spielman A, Matuschka FR. Adaptation of diverse Lyme disease spirochetes in a natural rodent reservoir host. Infect Immun. 2004; 72(4):2442–4. https://doi.org/10.1128/IAI.72.4.2442-2444.2004 WOS:000220481600076. PMID: 15039378
- Gatewood AG, Liebman KA, Vourc'h G, Bunikis J, Hamer SA, Cortinas R, et al. Climate and tick seasonality are predictors of *Borrelia burgdorferi* genotype distribution. Appl Environ Microbiol. 2009; 75 (8):2476–83. https://doi.org/10.1128/aem.02633-08 ISI:000264936800028. PMID: 19251900